**Author’s response to reviews**

**Title:** A Rare Cause of Small Bowel Obstruction, Phytobezoar in a Jejunal Diverticulum: A case report

**Authors:**

mohammad tayeb (drmtayeb@yahoo.com)
faiz khan (faizkhan.surg@yahoo.com)
fozia rauf (farauf@yahoo.com)
mumtaz khan (mmutazkhan@gmail.com)

**Version:** 4  **Date:** 15 May 2011

**Author’s response to reviews:**

Cover letter
To
The JMCR Editorial Team
Subject: Making the suggested changes by editorial team in our Article
Case report: A Rare Cause of Small Bowel Obstruction, Phytobezoar in a Jejunal Diverticulum: A case report.
Authors:
mohammad tayeb (drmtayeb@yahoo.com)
faiz khan (faizkhan.surg@yahoo.com)
fozia rauf (farauf@yahoo.com)
mumtaz khan (mmutazkhan@gmail.com)
MS: 1624372875499765
The editorial team suggested the following changes which I did.

*To change the title and we did that according to your suggestion.

*Now in this revised version I have included age and ethnicity of patient in the Abstract.

*The editorial team have quite rightly pointed out that 'flood affected IDP (internally displaced person) from Charssada', should be removed as this is an identifier, and is not immediately relevant to the Case Report and so we did that as well.

*I am uploading all figures separately as figure files on the submission system in this revised manuscript.

I am grateful to the editorial team for their quick response, valuable suggestions and help regarding publication of this article.
With Best Wishes
Mohammad Tayeb